Author's response to reviews

Title: A draft framework for measuring progress towards the development of a national health information infrastructure

Authors:

Dean F. Sittig (dean.f.sittig@kp.org)
Richard N Shiffman (richard.shiffman@yale.edu)
Kevin Leonard (k.leonard@utoronto.ca)
Charles Friedman (friedmc1@mail.nlm.nih.gov)
Barbara Rudolph (brudolph@leapfroggroup.org)
George Hripcsak (hripcsak@columbia.edu)
Laura L Adams (ladams@riqi.org)
Lawrence C Kleinman (lkleinma@NEMOURS.ORG)
Rainu Kaushal (RKAUSHAL@PARTNERS.ORG)

Version: 4 Date: 18 May 2005

Author's response to reviews:

We have addressed all comments from the editor including:

1. fixed title
2. added countries to each affiliation
3. added sections to abstract
4. removed footnotes
5. added manuscript sections
6. clarified competing interests of 2 authors.
7. fixed figure legends to better reflect content of figures
8. cropped figures more closely.

thanks for the help.
dean